Dear Editor,

I thank you for encouraging us to revise our manuscript once more in the light of the comments. We did our best on having two other native English teachers (from England) to revise our English writing. I believe we followed every suggestion and comment to resubmit the manuscript once more to your appreciation. We are very much pleased for this new opportunity. Every modification in the text is in red color.

We also believe in the main purpose of this work that was to emphasize the need of an urgent program for the disinfection of the water purification systems and dialysis machines [14], in health care centers and in pharmaceutical industries, because water is the principal incipient used in pharmaceutical manufacturing processes. As we stated in the manuscript, "The water used in health center areas and in pharmaceutical industries should be periodically analyzed as a preventive measure against the spreading of microorganisms, allowing measures of improvement to be taken rapidly, as required".

Every suggestion made by the second reviewer was also followed. We also included in "Conclusions" the paper suggested.

Looking forward to receiving your appreciation,

Sincerely yours,

Thereza Christina Vessoni Penna